Author's response to reviews

Title: Sexual Attitudes, Pattern of Communication, and Sexual Behavior Among Unmarried Out-of-School Youth in China

Authors:

Bo Wang (bwang@med.wayne.edu)
Xiaoming Li (xiaoli@med.wayne.edu)
Bonita Stanton (BStanton@dmc.org)
Vafa Kamali (vafa.kamali@usm.edu)
Sylvie Naar-King (snaarkin@med.wayne.edu)
Iqbal Shah (shahi@who.int)
Ronald Thomas (rthomas@med.wayne.edu)

Version: 3 Date: 17 July 2007

Author's response to reviews: see over
July 17, 2007

The BioMed Central Editorial Team

Re: Revision Manuscript (2033772719137396)

Dear Editor:

Thank you for your email dated June 29, 2007. My collaborators and I read through the reviewer’s comments and your letter carefully. We agreed with the two recommended changes and made the suggested revisions.

Enclosed please find a copy of our revised manuscript entitled “Sexual Attitudes, Pattern of Communication, and Sexual Behavior Among Unmarried Out-of-School Youth in China” and our detailed response to the reviewer. Please let me know if you have any questions.

Thank you for your consideration of our manuscript.

Sincerely yours,

Bo Wang, Ph.D.
Department of Community Health Sciences
The University of Southern Mississippi
118 College Drive, Box 5122
Hattiesburg, MS 39406
Tel: (601)266-6507
Fax: (601) 266-5043
Email: wangboous@yahoo.com
**Response to Reviewer 2:**

Minor Essential Revisions

1. *On page 6, the authors state (line 8) that the study was designed to address three primary research questions and then list 4. The analyses do relate to the four questions listed, so the number three should be changed.*

   **Response:** We agree with the reviewer and have changed “three primary research questions” to “four primary research questions.” (See page 6, line 8).

2. *On that same page, line 12, insert "are" between factors that and associated.*

   **Response:** We agree with the reviewer and have made the suggested changes (See page 6, line 12).